2023, Vol. 59, No. 12, 2237–2247 https://doi.org/10.1037/dev0001632

# Pregnant Women's Autonomic Responses to an Infant Cry Predict Young Infants' Behavioral Avoidance During the Still-Face Paradigm

Bailey Speck<sup>1</sup>, Jennifer Isenhour<sup>1</sup>, Mengyu (Miranda) Gao<sup>2</sup>, Elisabeth Conradt<sup>1, 3</sup>,

Sheila E. Crowell<sup>1</sup>, and K. Lee Raby<sup>1</sup>

<sup>1</sup> Department of Psychology, University of Utah

<sup>2</sup> School of Psychology, Beijing Normal University

<sup>3</sup> Department of Psychiatry and Behavioral Sciences, Duke University School of Medicine

Research suggests that women's autonomic nervous system responses to infant cries capture processes that affect their parenting behaviors. The aim of this study was to build on prior work by testing whether pregnant women's autonomic responses to an unfamiliar infant crying also predict their infants' emerging regulation abilities. Participants included 97 women in their third trimester of pregnancy, located in the United States. Most participants identified as White/non-Hispanic (48%) or Hispanic (30%), their mean age was approximately 30 years, and the modal family income was \$40,000-\$79,999. Pregnant women's respiratory sinus arrhythmia (RSA) and skin conductance levels (SCL)-which are thought to capture emotional engagement and behavioral inhibition, respectively-were measured while the women watched a relaxing video and a video of an unfamiliar infant crying. Approximately 7 months later, women and their infants completed the still-face paradigm (SFP). Infants' avoidance and resistance behaviors during the SFP reunions were rated. Pregnant women's RSA and SCL responses to the infant cry video uniquely predicted infants' avoidance (but not resistance) during the SFP. Infants displayed higher levels of avoidance when their mothers exhibited lower levels of RSA reactivity or when their mothers exhibited higher levels of SCL activity in response to the infant cry video. Maternal sensitivity during mother-infant free-play interactions did not mediate the associations between pregnant women's autonomic responses to the cry video and infants' avoidant behavior during the SFP. Discussion focuses on potential mechanisms underlying associations between pregnant women's autonomic responses to infant distress and infants' socioemotional development.

#### Public Significance Statement

Autonomic markers of pregnant women's emotional responses to an unfamiliar infant crying collected before the birth of their child could predict their young infants' later socioemotional behaviors. Because infant development provides a foundation for later social and emotional development, these findings suggest that interventions targeting pregnant women's emotional responses to infants' distress cues may lead to improvements in children's socioemotional outcomes.

*Keywords:* infant cry, respiratory sinus arrhythmia, electrodermal activity, still-face paradigm, maternal sensitivity

Supplemental materials: https://doi.org/10.1037/dev0001632.supp

Crying is a social communication signal commonly expressed by infants during the first year of life (Esposito & Bornstein, 2019; Vermillet et al., 2022). Infants cry when they feel discomfort, and the cry serves the purpose of eliciting a parental response (Bell & Ainsworth, 1972; Bowlby, 1969/1982). Parents' behavioral responses to infants' cries and other distress signals are theoretically expected to

Editor's Note. Koraly Pérez-Edgar served as the action editor for this article.—KP-E

This article was published Online First September 28, 2023.

Bailey Speck D https://orcid.org/0000-0001-5569-1310

This study was supported by grants from the National Institute of Mental Health (R01MH119070 and R21MH109777). Any opinions, findings, conclusions, or recommendations expressed in this material are those of the authors and do not necessarily reflect the views of the National Institutes of Health. Preregistration materials for this article can be found at https://osf.io/emfp5/.

Bailey Speck served as lead for conceptualization, formal analysis, and

writing-original draft. Jennifer Isenhour served in a supporting role for resources and writing-review and editing. Mengyu (Miranda) Gao served in a supporting role for writing-review and editing. Elisabeth Conradt served in a supporting role for funding acquisition, resources, and writing-review and editing. Sheila E. Crowell served in a supporting role for funding acquisition, resources, and writing-review and editing. K. Lee Raby served as lead for supervision and served in a supporting role for conceptualization, formal analysis, writing-original draft, and writing-review and editing.

Correspondence concerning this article should be addressed to Bailey Speck, Department of Psychology, University of Utah, 380 South 1530 East, Behavioral Sciences Building Room 506, Salt Lake City, UT 84112, United States. Email: bailey.speck@psych.utah.edu shape children's socioemotional development by scaffolding the development of self-regulation and contributing to a positive representation of the self and others (Leerkes et al., 2009; Sroufe, 1997; Thompson, 1997). Indeed, empirical data indicate that parental sensitivity in response to infant distress predicts infant–parent attachment security, child social competence, and fewer child behavior problems (Bornstein & Tamis-Lemonda, 1997; Leerkes, 2011; Leerkes & Zhou, 2018; Leerkes et al., 2009, 2011; McElwain & Booth-LaForce, 2006; Wright et al., 2018).

Parents' autonomic nervous system responses to their infants' distress are also associated with infants' early socioemotional development (Groh et al., 2019; Hill-Soderlund et al., 2008; Leerkes et al., 2017). For example, Groh et al. (2019) reported that a lesser degree of maternal respiratory sinus arrhythmia (RSA) withdrawal during a distressing caregiving task when infants were 6 months old predicted greater infant avoidance when infants were 12 months old. One possible explanation for these findings is that parent's autonomic responses to infant cries capture psychological processes that shape how parents interact with their infants and therefore their infants' development. An alternative possibility is that infants elicit different autonomic responses from their parents. In the current study, we attempt to address this issue by assessing whether pregnant women's autonomic responses to unfamiliar infant cries predict their infant's later regulation abilities. Autonomic responses in this context are thought to reflect emotional responding to infant distress cues, and are expected to affect infant regulation indirectly via differences in parenting behaviors. Capturing parents' emotional responding prior to the birth of the child will clarify the role of parents' autonomic responses in potentially shaping children's developmental outcomes. In addition, a better understanding of the predictive power of maternal responses to infant cry signals during the prenatal period will help with earlier identification of risk factors for problems related to infants' regulatory outcomes even before the child is born.

## Adults' Autonomic Responses to Infant Cries

Measures of individuals' autonomic nervous system responses to stimuli are commonly used to capture cognitive and emotional reactions that may be automatic and less susceptible to response biases than are self-report questionnaires and behavioral observations (Dennis et al., 2012; Raby et al., 2015). One commonly investigated measure of parasympathetic nervous system activity is RSA. RSA refers to the heart rate accelerations and decelerations as a function of respiration via the vagus nerve (Shaffer et al., 2014). According to polyvagal theory (Porges, 2007), engagement in social relationships is supported by increased vagal influence on the heart. From this perspective, decreases in RSA in response to a challenging interpersonal situation reflect emotional engagement and coping with the demands of the social task. Likewise, a lack of RSA decreases in these situations is thought to reflect difficulties in social and emotional regulation (Porges, 2007). It is important to note that the implications of RSA reactivity appear to vary based on the nature of the task (Beauchaine et al., 2019). However, within the context of caring for a young child, decreases in RSA are generally thought to facilitate a parent's capacity to sensitively respond to their child's emotional signals (e.g., Mills-Koonce et al., 2007; Moore et al., 2009; Xu & Groh, 2023).

A growing number of studies appear to support the idea that RSA reductions in response to infant distress reflect psychological processes that support responsive caregiving. For example, blunted maternal RSA reductions in response to their own infants' displays of negative emotions have been associated with less maternal sensitivity (Mills-Koonce et al., 2007; Moore et al., 2009; cf. Leerkes et al., 2017). Likewise, pregnant women who exhibit minimal changes in RSA levels in response to audio- or video-recorded cries of unfamiliar infants tend to report more negative perceptions of infants' cries and lower levels of observed maternal sensitivity (Ablow et al., 2013; Joosen, Mesman, Bakermans-Kranenburg, Pieper, et al., 2013; Leerkes, Su, et al., 2016). To the best of our knowledge, there are no data that relate pregnant women's RSA responses to an unfamiliar infant cry and their infants' later social and emotional outcomes. However, a lack of RSA decreases while interacting with one's own distressed infant is associated with infant attachment insecurity (Groh et al., 2019; Hill-Soderlund et al., 2008).

A second commonly investigated measure of parents' autonomic nervous system activity is skin conductance levels (SCL). SCL is driven by sympathetic nervous system arousal (Dawson et al., 2017), and increases in SCL in response to a stressful task may reflect attempts to inhibit emotional expressions or behavior (Fowles, 1980, 1988). Based on this perspective, increases in SCL in response to infant crying may be associated with less sensitive parental responses to infant distress, which in turn will fail to support young children's emerging abilities to effectively regulate their own emotions and behaviors. Indeed, SCL increases in response to audio-recordings of an unfamiliar infant crying have been associated with more negative emotional reactions to the infant cry, more harsh discipline practices, and less sensitive parenting (Emery et al., 2014; Joosen, Mesman, Bakermans-Kranenburg, & van IJzendoorn, 2013), although null findings have also been reported (Leerkes et al., 2023, Leerkes, Gedaly, & Su, 2016). Although these findings may suggest that increases in SCL in response to infant cries are associated with parenting-related emotions and behaviors, we are not aware of any studies testing whether increases in SCL in response to unfamiliar infants' crying are associated with infants' later socioemotional outcomes.

In considering to consider potential main effects of RSA and SCL responses to unfamiliar infant cries, Leerkes et al., (Leerkes et al., 2015; Leerkes, Gedaly, & Su, 2016) have proposed that parents' RSA and SCL responses to infant distress signals may interact with one another to predict parent and infant outcomes. According to this perspective, it is the combination of both high physiological arousal (as indexed by SCL increases) and high physiological regulation (as reflected by RSA decreases) in response to infant distress that supports appropriate parental responses. Indeed, RSA and SCL responses to unfamiliar infant cries interacted in this manner to predict mothers' empathetic assessments of the unfamiliar infant's cry as well as maternal sensitivity when interacting with their own infants (Leerkes et al., 2015, 2023; Leerkes, Gedaly, & Su, 2016). Moreover, infants were less likely to develop secure attachments and more likely to develop a disorganized attachment if their mothers exhibited both high SCL reactivity and a lack of RSA declines in response to their own infants' distress (Leerkes et al., 2017; Leerkes, Su, et al., 2016).

Altogether, these findings suggest that adults' autonomic responses to an unfamiliar infant's crying reflect emotional processes that may contribute to how they will ultimately interact with their own infants. It remains unclear, however, if adults' autonomic responses to videos of unfamiliar infants also shape infants' socioemotional development. As previously noted, existing research in this area has focused on parents' autonomic responses during parent–infant interactions (e.g., Groh et al., 2019; Hill-Soderlund et al., 2008; Leerkes et al., 2017). However, given the transactional nature of parent-child interactions, it is possible that parents' autonomic responses in these contexts may be due to child-driven effects on parents' autonomic responses (e.g., DePasquale, 2020; Gao et al., 2022; Ostlund et al., 2017). Controlling for infant characteristics in analyses is an imperfect solution due to measurement error and the inability to assess all relevant infant characteristics. The present study more directly addresses this issue by testing whether women's responses to infant distress while they were pregnant predicted their infants' socioemotional outcomes at age 7 months. By assessing women's autonomic responses to infant cry stimuli prior to the birth of their child, this approach eliminates the possibility that variability in parents' emotional responses during the cry are due to individual differences in their own infant's characteristics or behavior. As previously noted, we are not aware of any studies that have utilized this type longitudinal design to assess whether women's autonomic responses to infant cry stimuli while they are pregnant predict later infant outcomes. Addressing this gap will help us clarify the predictive nature of parents' responses to infant distress.

## Infant Behavior During the Still-Face Paradigm

The still-face paradigm (SFP; Tronick et al., 1978) is a wellvalidated, structured laboratory task for assessing infants' early socioemotional development. During the SFP, the parent is first instructed to first sit across from their infant and play with them as they normally would for 2 min. Then, the parent is asked to display a neutral facial expression, remain still, and not respond to the infant for 2 min. Most infants respond to the psychological separation from the parent during the still-face episode with decreases in positive affect and increases in negative affect (Mesman et al., 2009). Finally, the parent is instructed to resume normal interaction with the infant for an additional 2 min.

During the reunion episode of the SFP, infants, and parents engage in a mutually regulated process to repair the disruption in the interaction and reestablish harmonious, back-and-forth interactions. Parents typically comfort their infants and encourage them to resume play interactions (e.g., Mesman et al., 2009, 2013), but infants are also engaging in self-regulation during the reunion episodes of the SFP. Kogan and Carter (1996) created a system for coding these selfregulation behaviors and strategies during these distressing parentchild interactions. The attention maintenance rating captures the degree to which infants look at, reach toward, and respond to their parents with positive affect during the SFP reunion. Alternatively, the avoidance rating measures the degree to which infants withdraw from the parent and actively disengage from interactions, and the resistance rating reflects the degree to which infants display high levels of distress and anger in response to the parent's attempts to interact with them (Kogan & Carter, 1996). These three behavioral rating scales are thought to capture strategies for self-regulation that are shaped by prior parent-infant interactions. The ratings of infant avoidance and resistance during the SFP reunion episodes, in particular, are thought to represent maladaptive self-regulation strategies that develop in the context of insensitive care. Indeed, infants' self-regulation behaviors during the SFP reunion episodes are associated with parents' caregiving behaviors (Conradt & Ablow, 2010; Kogan & Carter, 1996). These early emerging strategies for regulating one's behavior during distressing parent-child interactions have been shown to predict later attachment behaviors during the Strange Situation (Kogan & Carter, 1996; see also Barbosa et al., 2021).

#### **Current Study**

The primary aim of the current study was to test whether pregnant women's autonomic responses to a video of an unfamiliar infant crying predict their young infants' later self-regulation behaviors during the SFP. This study is unique because infants' self-regulation strategies were observed during the reunion episodes of the SFP approximately 7 months after the assessment of parents' autonomic responses. To the best of our knowledge, this is the first study to examine the potential intergenerational associations between pregnant women's autonomic nervous system responses to infant cries and later infant socioemotional outcomes.

Our first hypothesis was that minimal RSA reductions in response to the infant cry would predict infants' use of maladaptive regulation strategies (i.e., avoidance and/or resistance) during the SFP. This is based on the idea that a lack of RSA reductions reflects emotional disengagement with the task (Porges, 2007) as well as evidence that a lack of RSA reductions to infant cry stimuli may be associated with insensitive parenting (e.g., Ablow et al., 2013; Joosen, Mesman, Bakermans-Kranenburg, Pieper, et al., 2013; Mills-Koonce et al., 2007; Moore et al., 2009). Our second hypothesis was that increases in pregnant women's SCL in response to the infant cry would predict their infants' use of behavioral avoidance and resistance during the SFP. This is based on the idea that SCL increases reflect emotional inhibition (Fowles, 1988) as well as evidence that SCL increases in response to infant cry stimuli may be associated with negative emotional reactions to the cry and harsh parenting (e.g., Emery et al., 2014; Joosen, Mesman, Bakermans-Kranenburg, & van IJzendoorn, 2013). Based on the evidence from Leerkes et al., (Leerkes et al., 2023; Leerkes, Gedaly, & Su, 2016), we also explored the possibility that pregnant women's RSA and SCL responses to an unfamiliar infant's cry interacted with one another to predict infant avoidant and resistant regulation strategies.

A secondary aim of this study was to test whether associations between pregnant women's RSA and SCL responses to infant cry stimuli and their infants' later self-regulation behaviors were mediated by observed maternal sensitivity. We expected that pregnant women who exhibited minimal reductions in RSA or greater increases in SCL activity in response to the infant cry stimuli would interact with their infants in a less sensitive manner, and this in turn would be associated with infants' avoidance and resistance behaviors during the reunion episode of the SFP.

## Method

#### **Participants**

This study used data from a longitudinal study of 162 women and their children. Pregnant women were recruited through flyers, brochures, and social media posts. Because the area in which the data were collected consists of predominantly non-Hispanic, White individuals, women of color were intentionally oversampled in an effort to maximize the likelihood that the findings from this study generalize to individuals that are more racially and ethnically diverse than the demographics of our study location. In addition, women with high and low scores on the Difficulties in Emotion Regulation Scale (Gratz & Roemer, 2004) were intentionally oversampled to achieve a uniform distribution on levels of emotion dysregulation. This is in alignment with the overarching aims of the study to understand the processes underlying intergenerational transmission of emotion dysregulation. Additional eligibility criteria for the study included participants being ages 18-40 at the time of their pregnancy, no reported pregnancy complications such as preeclampsia or gestational diabetes, no reported substance use during pregnancy, anticipated delivery of a singleton, and planned delivery at a participating hospital. Individuals interested in participating in the study first completed a screener questionnaire to determine their eligibility for the study, and eligible individuals were contacted to complete additional questionnaires and laboratory visits. Eligible women participated in a laboratory visit during their third trimester of pregnancy and then returned to the lab when their infants were approximately 7 months old (average age = 6.6 months). This time point was selected because this is the age that the SFP is most commonly conducted (Jones-Mason et al., 2018; Mesman et al., 2009). All mothers completed informed consent prior each visit, and this study was approved by the University of Utah's Institutional Review Board. In addition, individuals were compensated \$50 for their participation in the prenatal visit and \$50 for their participation in the 7-month visit. Mothers were also provided with a list of local and national resources relevant to the transition to parenthood.

Because this specific study is focused on infant self-regulation behaviors during the SFP reunion, mother-infant dyads were included in the analyses if ratings of infant behavioral self-regulation during the 7-month visit were available. Forty-eight dyads were excluded because they did not participate in the 7-month lab visit (29 indicated they were too busy to participate, 13 could not be contacted, four withdrew from the study, one participant moved, and one experienced a fetal demise). Of the remaining 114 participants, 17 were excluded because it was not possible to rate the infant's self-regulation behaviors during the lab visit (10 due to video error, three mothers blocked the view of the infant, two infants were too distressed to complete reunion episodes, and two mothers deviated from protocol, such as providing the infant with a pacifier). This resulted in a sample of 97 mother-infant dyads. Table 1 contains demographic information for these dyads. The median income for this sample of 97 mothers was comparable to the median income in the Salt Lake County area (\$82,206; U.S. Census Bureau, 2021). Demographic variables (e.g., maternal age, educational attainment, and race/ethnicity) were significantly associated with attrition from the prenatal to the 7-month visit, but none of these variables were significantly correlated with missing the ratings of infant self-regulation behaviors at the 7-month visit. With a sample of 97 dyads, the analyses had 80% statistical power to detect an effect of r = .28.

# Procedure

At the prenatal lab visit, women were fitted with electrodes used to collect electrocardiogram and skin conductance data. They completed a 10-min resting baseline, followed by the Trier Social Stress Test (Kirschbaum et al., 1993) and a 9-min resting recovery period. This extended resting recovery period was intended to allow women's physiological responses to recover after the Trier Social Stress Test. After the recovery period, a research assistant returned to the room and told the women they would be watching a series of video stimuli. The women were not given any additional instructions. The women watched a 1-min seascape video baseline intended to elicit a relaxed state. They then watched a 1-min video of an infant girl playing with an adult woman, followed by a 1-min video of the infant girl sitting alone on the floor crying. Finally, the women

#### Table 1

Demographic Characteristics of Mothers and Infants at the 7-Month Time Point

Demographic characteristic	Mothers	Infants
Average age (SD)	29.7 years (4.6)	6.6 months (0.5)
Sex assigned at birth	•	
Female (%)	100.0	47.4
Male (%)	0.0	52.6
Race/ethnicity		
White (%)	47.9	50.5
Hispanic (%)	30.3	26.8
Multiracial (%)	10.4	16.5
Asian (%)	9.4	3.1
Black (%)	1.0	1.0
Other (%)	1.0	2.0
First-time parent (%)	48.9	_
Relationship status		
Married (%)	76.0	_
Single and never married (%)	17.7	_
Separated or divorced (%)	6.3	_
Educational attainment		
Less than high school degree (%)	2.1	_
High school degree or GED (%)	13.7	_
Some college (%)	31.6	_
Baccalaureate degree (%)	31.6	_
Postbaccalaureate degree (%)	21.1	_
Household income		
<\$9,000 (%)	4.2	_
\$9,000-\$14,999 (%)	4.2	_
\$15,000-\$24,999 (%)	11.4	_
\$25,000-\$39,000 (%)	12.5	_
\$40,000-\$79,999 (%)	37.5	_
\$80,000-\$99,999 (%)	13.5	_
>\$100,000 (%)	12.5	_

*Note.* GED = general educational development.

watched an additional seascape video intended to serve as a recovery period.

At the 7-month visit, mothers and their infants were fitted with electrodes. Mothers and infants watched a 2-min Baby Einstein baseline video, which was intended to induce a neutral and calm state, and mothers then completed a brief writing task. Afterward, mother-infant dyads participated in the extended version of the SFP. This task consisted of a 2-min play interaction where dyads were seated across from one another, followed by a 2-min still-face episode where the mother was instructed to remain affectively neutral, followed by a 2-min reunion episode where the mother was instructed to reengage the infant in play. In the extended SFP procedure, there was a second 2-min still-face and 2-min reunion episode immediately following the first reunion. This five-episode task was selected because it elicits a stronger emotional response from infants than the traditional three-episode still-face (Haley & Stansbury, 2003; Jones-Mason et al., 2018). After the SFP, electrodes were removed and the mother-infant dyads completed a free-play interaction.

# Measures

#### Maternal RSA

At the prenatal lab visit, electrocardiograph data were continuously collected during all four videos using a standard three-lead spot electrode placement on the women's torsos (i.e., right clavicle and lowest left rib). The data were sampled at 500 Hz using MindWare Technology mobile devices (MindWare Technologies Ltd.; Biolab software Version 3.1), and electrocardiograph activity was monitored throughout the task to ensure accurate data collection. Later, MindWare Technology's heart rate variability analysis was used for scoring the data. MindWare's software automatically identified R peaks within the QRS complex, which were visually inspected and corrected if needed by trained research assistants. Using MindWare's software, spectral analysis was used to decompose the variability in beat-to-beat intervals in 60-s intervals into very low-frequency (0.003–0.04 Hz), low-frequency (0.04–0.12 Hz), and high-frequency power (0.12-0.60 Hz; MindWare Technologies, 2014). RSA was calculated as the natural log of the high-frequency power spectrum. After data were cleaned and scored, boxplots were run to check for outliers, which were flagged as needed. Infants whose data were flagged were examined an additional time to check for scoring errors, which were corrected if identified.

#### Maternal Electrodermal Activity

At the prenatal lab visit, SCL data were continuously collected during all four videos by placing two skin conductance electrodes with 0.5% NaCl solution to the thenar and hypothenar eminences of each participant's nondominant hand. The data were sampled at 500 Hz using MindWare Technology's mobile devices (MindWare Technologies Ltd.; Biolab software Version 3.1). The SCL data were monitored throughout the task to ensure accurate data collection. Later, MindWare Technology's electrodermal activity analysis was used for scoring the data. MindWare's software automatically flagged peaks and troughs in the electrodermal signals and these flagged peaks and troughs were visually inspected by trained research assistants. Using MindWare's software, SCL was calculated by averaging the SCL, in microsiemens, across each 30-s epoch. The data were manually reviewed for outliers, and participants whose data were flagged were examined an additional time to check for scoring errors, which were corrected if identified.

#### Infant Regulation Behaviors

Infant regulation behaviors during the two reunion episodes of the SFP were rated using the coding system designed by Kogan and Carter (1996). During each reunion episode, infants' attention maintenance, avoidance, and resistance were rated using a set of 3-point scales. Examples of attention maintenance included eye contact with the mother, reaching for the mother, and responsiveness to the maternal bids for engagement. Examples of infant avoidance included gaze aversion, attempting to turn the body away from the mother, and delayed or no response to maternal bids for attention. Examples of infant resistance included continued and persistent fussing, protesting while looking at the mother, and angrily pulling away from the mother's touch. Previous studies using this coding system have reported that these ratings are concurrently correlated with maternal behaviors during the SFP (Conradt & Ablow, 2010; Kogan & Carter, 1996; Rosenblum et al., 2002) and predict infant attachment classifications during the Strange Situation Procedure (Kogan & Carter, 1996).

Twenty-four percent of cases were double-coded, and the intraclass correlations were between .93 and .97. An exploratory factor analysis (principal axis factoring, oblique rotation) of the six ratings of infants' behaviors during the two reunion episodes of the SFP identified two factors. The first factor included the ratings of infant avoidance during the two reunions as well as infant attention maintenance during the first reunion. The second factor included the two ratings of infant resistance as well as infant attention maintenance during the second reunion. Because infant attention maintenance ratings cross-loaded on the two factors, they were excluded from analyses. Infant avoidance was calculated by averaging the ratings of avoidance during the two reunions (r = .56, p < .001), and infant resistance was calculated by averaging the ratings of infant resistance during the two reunions (r = .67, p < .001).

## Maternal Sensitivity

Mothers' behavior during the two, 2-min reunion episodes of the SFP were coded using an adapted version of the Observational Record of the Caregiving Environment (NICHD Early Child Care Research Network, 1996). During both episodes, mothers' behaviors were rated using five rating scales: maternal sensitivity to infant distress, maternal sensitivity to infant nondistress, intrusiveness, positive regard, and detachment. All ratings were assigned using a 5-point scale (1 = not at all characteristic to 5 = highly characteristic). All observations were double-coded by research assistants, and disagreements in coding decisions greater than a half point were conferenced. Intraclass correlation estimates for the 10 sets of ratings were between .42 and .80 (average intraclass correlation coefficient = .73). For the purposes of this study, the ratings of maternal sensitivity to infant distress and nondistress were used. Indicators of maternal sensitivity to infant distress include the proportion of distress signals responded to, the latency of response, and the appropriateness of response. Maternal sensitivity to nondistress was assigned when infants did not display overt distress during the reunion portion of the SFP, and was defined as following the child's lead during the parent-infant interaction. This rating was considered conceptually relevant because it was assigned during the reunion episode of the SFP, which is potentially distressing context. Maternal sensitivity to infant nondistress was significantly correlated with maternal sensitivity to infant distress during both reunion episodes (r = .46, p = .043 and r = .54, p = .009). Therefore, the four ratings of maternal sensitivity during the two SFP reunion episodes were averaged to create an overall score of maternal sensitivity in the context of infant distress  $(\alpha = .74).$ 

## **Covariates**

Five covariates were considered for all analyses, including parity  $(1 = first-time \ parent, \ 0 = otherwise)$ , family socioeconomic status, which was calculated by standardizing and averaging measures of women's household income, educational attainment, and household occupational prestige (Hout et al., 2016;  $\alpha = .79$ ), infant race/ethnicity, infant biological sex (1 = male, 2 = female), and infant age in months at the time of the SFP. These covariates were considered because they have been reported to be associated with parents' autonomic responses to infant distress as well as infants' socioemotional development (e.g., Groh et al., 2019; Leerkes, Su, et al., 2016). Preliminary analyses indicated that only infant race/ethnicity was significantly correlated with either the measures of mothers' RSA or SCL responses to the infant cry video or the measures of infant regulation behaviors during the SFP. Therefore, only infant race/ethnicity ultimately was included as a covariate in our analyses. This was represented with two binary variables: one representing whether infant was Hispanic or not (the most common marginalized group in our sample) and the other indicating whether the infant was a non-Hispanic person of color (e.g., Black/ African American, Asian American, or multiracial) or not.

## Analytic Plan

Prior to the main analyses, a series of paired-samples *t* tests were conducted to assess whether pregnant women's average RSA and SCL levels changed across the videos. Then, a measure of RSA reactivity to the infant cry video was created by subtracting mothers' average RSA levels during the video baseline from their average levels during the infant cry video. Similarly, a measure of SCL reactivity to the cry video baseline from their average SCL during the video baseline from their average SCL during the video baseline from their average SCL during the video baseline from their average levels during the infant cry video. Next, we examined the bivariate correlations among the measures capturing women's autonomic responses to the infant videos, maternal sensitivity, the potential covariates, and infant regulation behaviors.

In order to test whether pregnant women's autonomic responses to a video of an unfamiliar infant crying predict young infants' selfregulation behaviors, we conducted two hierarchical linear regression analyses. One focused on predicting infant avoidance behaviors during the SFP, and the other predicted infant resistance behaviors. For each model, maternal RSA and SCL reactivity to the infant crying videos were included as predictor variables along with covariates.

The indirect effects of maternal RSA and SCL reactivity to the infant cry on infants' regulation behaviors via maternal sensitivity during the SFP reunion were tested in Mplus. Maternal RSA and SCL reactivity to the infant crying video were included as predictors, along with the covariates, and maternal sensitivity in response to infant distress was entered as a mediator. Additional preregistered analyses, which assess maternal sensitivity in a free-play context as a potential mediator of the association between women's autonomic responses and infants' regulatory behaviors, are reported in the online supplemental materials of this article.

## Missing Data

Of the 97 dyads included, 4% of mothers were missing SCL data during the baseline video, 6% were missing SCL data during the play video, 2% were missing SCL during the cry video, and 1% were missing SCL data during the recovery video. Additionally, 3% of mothers were missing RSA data during the baseline video, 3% were missing RSA data during the play video, 1% were missing RSA during the cry video, and 2% were missing RSA data during the recovery video. Approximately 4% of infants were missing avoidance data but not resistance, and 1% were missing resistance data but not avoidance. No mothers were missing maternal sensitivity data. Missing data were handled using Full Information Maximum Likelihood in Mplus (Muthén & Muthén, 2017) for the bivariate correlations and focal analyses. The overall research design and hypotheses for this study were preregistered with the Open Science Framework (Speck et al., 2023). The current study differs from the initial preregistration with regard to the variables used for covariates and mediators, and these differences have been highlighted. Hypotheses and methods pertaining to the main effects of mothers' autonomic responses to the video of an unfamiliar infant crying and their own infants' self-regulatory behavior during the SFP were consistent with the initial preregistration. Data and code for this study are not publicly available online but can be made available upon request.

#### Results

## **Preliminary Analyses**

On average, RSA decreased from the seascape baseline video to the video of the mother–infant playing, t(93) = -6.18, p < .001, but did not significantly change from the mother–infant play video to the infant cry video, t(93) = -1.0, p = .32, or from infant cry video to the recovery video, t(94) = 0.46, p = .64; Figure 1.

SCL decreased from baseline to the mother–infant play video, t(91) = -7.06, p < .001, but then increased from the mother–infant play video to the infant cry video, t(91) = 2.34, p = .02. SCL decreased from the infant cry video to the recovery video, t(94) = -3.81, p < .001; Figure 2.

Descriptive statistics and correlations among study variables are presented in Table 2. In addition, there were no outlier values for any of the variables and skewness statistics were less than 0.92 for variables used in the analyses.

# Do Maternal Autonomic Responses to the Infant Cry Predict Infant Regulatory Behaviors During the SFP?

Ordinary least squares regressions are presented in Table 3. Pregnant women's RSA reactivity to the cry video was positively associated with infant avoidance during the reunion episode of SFP. In other words, infants tended to display higher levels of avoidance when their mothers exhibited lower RSA reactivity to the task (i.e., had RSA reactivity values close to zero as opposed to a larger negative value). Women's SCL reactivity to the cry video was also positively associated with infant avoidance, which suggests that infants displayed higher levels of avoidance when their mothers exhibited higher SCL in response to the infant cry video. Neither RSA reactivity nor SCL reactivity was significantly associated with infant resistance.

A set of exploratory analyses were conducted to determine whether maternal RSA and SCL reactivity to the unfamiliar infant

#### Figure 1

Pregnant Women's Average RSA Levels Across the Video Stimuli



*Note.* Error bars represent 1 standard error. RSA = respiratory sinus arrhythmia. \*\* p < .01.

Figure 2 Pregnant Women's Average SCL Across the Video Stimuli



Note. Error bars represent 1 standard error. SCL = skin conductance levels. p < .05. p < .01.

cry video interacted to predict infant regulation behaviors. The maternal RSA and SCL reactivity variables were mean-centered and then multiplied together to create an interaction term. This interaction term was included as a predictor variable along with the centered RSA and SCL reactivity variables and covariates. Results indicated that maternal RSA and SCL reactivity did not interact to predict infant avoidance behavior or resistance behavior ( $\beta = .08$ , SE = 0.11, p = .45, and  $\beta = .06$ , SE = 0.12, p = .60, respectively).

In order to test whether these associations were unique to mothers? autonomic responses to the infant cry video, measures of maternal RSA and SCL reactivity to the infant play video were calculated by subtracting mothers' average autonomic levels during the video baseline from their average levels during the infant play video. Two additional hierarchical linear regression analyses were conducted to test whether maternal autonomic responses to the infant play video predict young infants' self-regulation behaviors during the SFP. Once again, maternal RSA and SCL reactivity to the infant play videos were included as predictor variables along with covariates. Results indicated that neither pregnant women's RSA reactivity nor their SCL reactivity to the mother-infant play video significantly predicted infants' avoidance or resistance behaviors (p values between .15 and .55).

Table 2	2
---------	---

Descriptive Statistics and Correlations Among variable	Descriptive	<i>Statistics</i>	and (	Correlations	Among	Variable
--	-------------	-------------------	-------	--------------	-------	----------

# **Does Maternal Sensitivity Mediate the Associations Between Maternal Autonomic Reactivity and Infant** Avoidance?

We tested whether maternal RSA and SCL reactivity were indirectly associated with infants' avoidance behaviors via maternal sensitivity (see Table 4). Neither RSA reactivity nor SCL reactivity to the infant cry were significantly associated with maternal sensitivity. RSA reactivity and SCL reactivity also did not have significant indirect effects on infant avoidance. Moreover, SCL reactivity continued to predict infant avoidance behaviors when maternal sensitivity was included as a potential mediator, and RSA reactivity became marginally significant when maternal sensitivity was included as a potential mediator.

## Discussion

The primary aim of this longitudinal study was to test whether pregnant women's RSA and SCL responses to a video of an unfamiliar infant crying predict their infants' regulation behaviors during the reunion episode of the SFP. Our findings partially supported our hypotheses. Specifically, lower levels of maternal RSA reactivity and higher maternal SCL reactivity in response to the infant cry video were both uniquely associated with high levels of infant avoidance during the reunion episodes. These associations were specific to the video of an unfamiliar infant crying, as pregnant women's RSA and SCL responses to the video of the mother and infant playing did not significantly predict levels of infant avoidance. These findings extend earlier evidence indicating that parents' autonomic responses during parent-infant interactions are associated with parental sensitivity and infant behaviors (Groh et al., 2019; Hill-Soderlund et al., 2008; Mills-Koonce et al., 2007; Moore et al., 2009). To the best of our knowledge, this is the first study to provide evidence for an intergenerational association between pregnant mothers' RSA and SCL responses to an unfamiliar infant crying and their own infants' regulation behaviors.

Although we hypothesized that maternal responses would be associated with both infant avoidance and resistance, pregnant women's autonomic responses to the infant cry video were only significantly associated with levels of infant avoidance. Similar findings have been observed when assessing parents' RSA responses during parent-infant interactions. Specifically, mothers of infants who are classified as avoidantly attached (as opposed to secure or resistant) exhibit a lack of RSA reductions during both the SFP and the strange situation

Variable	1	2	3	4	5	6	7
1. Maternal RSA reactivity to infant cry video							
2. Maternal SCL reactivity to infant cry video	.06	_					
3. Maternal sensitivity	.14	.02	_				
4. Infant avoidance behavior	.17	.22*	.18	_			
5. Infant resistance behavior	01	.03	31**	.03			
6. Infant Hispanic versus non-Hispanic	07	09	10	.15	.00		
7. Infant non-Hispanic POC versus other	.25*	.12	.02	16	03	33**	_
M or %	-0.64	-0.78	3.09	1.11	1.39	27%	23%
SD	0.85	2.09	0.61	0.87	0.99		

*Note.* N = 97. RSA = respiratory sinus arrhythmia; SCL = skin conductance levels; POC = person of color. \* p < .05. \*\* p < .01.

Table 3

Predicting Young Infants' Regulatory Behaviors During the SFP From Pregnant Women's Autonomic Responses to an Infant Cry Video

	Outcon avo	me: int idance	Outcome: infant resistance			
Predictor variable	β	SE	р	β	SE	р
Maternal RSA reactivity	.21*	0.10	.04	00	0.11	.98
Maternal SCL reactivity	.25*	0.10	.01	.04	0.11	.74
Infant Hispanic versus non-Hispanic	.12	0.10	.25	00	0.11	.98
Infant non-Hispanic POC versus other	20	0.10	.05	03	0.11	.79

*Note.* N = 97. SFP = still-face paradigm; RSA = respiratory sinus arrhythmia; SCL = skin conductance levels; POC = person of color.

procedure (Groh et al., 2019; Hill-Soderlund et al., 2008). Consistent with polyvagal theory (Porges, 2007), minimal reductions in RSA in the context of infant crying may reflect a lack of emotional engagement with the infant's distress, and this in turn could lead to the type of dismissing and rejecting caregiving behaviors that are thought to contribute to the development of infant avoidance (Ainsworth et al., 1978). In addition, changes in SCL in response to infant distress may reflect the effortful inhibition of behavior (Dawson et al., 2017; Fowles, 1980), which may also lead to parental disengagement or rejection during parent–child interactions.

In addition to our test of the main effects of pregnant women's RSA and SCL responses to the infant cry, we also explored the possibility that pregnant women's RSA and SCL responses interacted with one another to predict infant regulation outcomes. However, our analyses did not provide support for the idea that RSA and SCL responses interact to predict infant avoidance or resistance. That said, it is important to interpret these null findings with caution because the current study was not sufficiently powered to test for moderation effects (Whisman & McClelland, 2005). Therefore, additional research involving larger samples is needed to test whether pregnant women's RSA and SCL responses to infant distress interact to predict infants' later socioemotional development.

The second aim of this study was testing whether the associations between pregnant women's RSA and SCL responses to an unfamiliar infant cry video and infants' regulation behaviors were mediated by maternal sensitivity. The results of these analyses indicated that the indirect effects of RSA and SCL reactivity to the infant cry and infant avoidance via maternal sensitivity were not statistically significant. There are at least two possible explanations for these nonsignificant results. First, the fact that maternal sensitivity and infant regulation behaviors were concurrently assessed in the same context may have resulted in distortions in the associations between those variables. Measures of maternal sensitivity assessed in other distressing contexts may be more appropriate to test this association. Similarly, more ecologically valid measures of maternal sensitivity in day-to-day parent-infant interactions may be more central in mediating the association between maternal autonomic responses and infant outcomes. A second explanation is that the associations between parents' autonomic responses to cry stimuli and their infants' regulation behaviors are not mediated by caregiving behavior. While maternal sensitivity to infant distress is thought to reflect processes relevant to children's self-regulation outcomes (Del Carmen et al., 1993; Lee et al., 2022; Leerkes et al., 2009), other mechanisms explaining the association between parents' autonomic responses and infant behaviors have been proposed. These include shared genetics, dyadic autonomic synchrony, and emotional contagion (Leerkes, Gedaly, & Su, 2016). Testing these potential mechanisms represents a valuable direction for future research.

One of the strengths of this article is that mothers' autonomic responses to infant cries were assessed prenatally. This aspect of the research design removes the possibility that infant characteristics are eliciting this parental physiological response. At the same time, this study does necessitate a reliance on the use of video clips of an unfamiliar infant crying, as opposed to one's own infant. This has the potential to weaken the associations between pregnant women's autonomic responses and later parenting or infant outcomes, given that parents' autonomic responses to an unfamiliar infant crying may not generalize to their own infant. The fact that the current study provided evidence for intergenerational associations between pregnant women's autonomic responses to an unfamiliar infant crying and their infants' regulation behaviors despite this potential limitation speaks to the significance of these autonomic responses.

A potential limitation of the study is that women's average SCLs unexpectedly decreased across the entire series of infant videos. This may have been due to the structure of the laboratory visit. Prior to watching the infant videos, the pregnant women completed the trier social stress task (TSST). Although there was a 9-min resting

#### Table 4

Indirect	Effects	of Maternal	Autonomic	Responses	on	Infant	Avoidance	<b>Behaviors</b>
During t	he SFP	via Materna	l Sensitivity					

Model	$\beta$ (SE)	р
Path A		
Maternal RSA reactivity $\rightarrow$ Maternal sensitivity	.14 (0.11)	.19
Maternal SCL reactivity $\rightarrow$ Maternal sensitivity	.01 (0.11)	.95
Path B		
Maternal sensitivity $\rightarrow$ Infant avoidance	.17 (0.10)	.08
Path C		
Maternal RSA reactivity $\rightarrow$ Infant avoidance	.19 (0.10)	.06
Maternal SCL reactivity $\rightarrow$ Infant avoidance	.24 (0.09)	.01
Indirect effects		
Maternal RSA reactivity $\rightarrow$ Maternal sensitivity $\rightarrow$ Infant avoidance	.02 (0.02)	.29
Maternal SCL reactivity $\rightarrow$ Maternal sensitivity $\rightarrow$ Infant avoidance	.00 (0.02)	.95

*Note.* N = 97. SFP = still-face paradigm; RSA = respiratory sinus arrhythmia; SCL = skin conductance levels.

<sup>\*</sup> p < .05.

recovery in between the TSST and the presentation of the infant videos, this task order may have nonetheless resulted in unusually highly SCLs during the baseline video administered immediately prior to the infant play and cry videos. That said, SCLs did significantly increase from the play to the cry video, and this was followed by a significant decrease from the cry to the recovery video. These changes across the tasks indicate that mothers were responding to the infant cry video with the anticipated increase in SCL, supporting the validity of our measure of SCL reactivity. Additionally, RSA levels exhibited the expected pattern of change across the videos. Specifically, RSA levels decreased from baseline to the infant play video with an additional (albeit nonsignificant) decrease during the infant cry video. Altogether this suggests that the videos were effective at eliciting autonomic responses from the pregnant women.

Future research may also benefit from the use of larger and more broadly representative samples. The current sample of 97 dyads included women who had relatively high levels of formal education, most of whom were married and had family incomes that were comparable to the median family income for the United States (Semega & Kollar, 2022). In addition, approximately half of the sample identified as White and non-Hispanic which is slightly lower than the United States (U.S. Census Bureau, 2022). Although the initial recruited sample was somewhat larger, attrition occurred between the prenatal and 7-month assessment largely due to mothers being too busy to participate after the birth of their child. Future research with larger samples will have more statistical power to precisely estimate the magnitude of the effect that women's autonomic responses have on infants' outcomes. Additionally, testing the presence of these associations in larger, more diverse, populations will help clarify how broader contextual characteristics, such as experiences of poverty or racial and ethnic discrimination, may affect the magnitude of the associations reported in the current study.

In the current study, the focal outcome was young infants' behaviors during the reunion episodes of the SFP, which is a well-validated task to assess young infants' emerging regulation strategies during parent-child contexts (Mesman et al., 2009). Parents' autonomic responses to an unfamiliar infants' cry may also predict infant mental health outcomes. Another fruitful direction for future studies may be to expand the scope of these initial findings to other infant socioemotional outcomes and potentially at later developmental time points. These research investments would also provide valuable information regarding opportunities for intervention.

In conclusion, the present longitudinal study provides novel evidence that lower levels of maternal RSA reactivity or higher levels of maternal SCL reactivity while pregnant and in response to an unfamiliar infant crying are associated with higher levels of infant avoidance behaviors during the SFP at age 7 months. To the best of our knowledge, this is the first known report of an intergenerational association between prenatal maternal autonomic responses to infant distress and later infant socioemotional outcomes. In this way, the results of this study suggest that parents' autonomic responses to infant distress capture emotional processes that may shape infant development. In addition to advancing our understanding the intergenerational effects of parents' emotional responding to infant cries, the results of this study can be applied toward developing novel interventions or modifying existing interventions that aim to alter infants' emerging regulation outcomes. Traditionally, interventions that focus on improving infants' social and emotional development have targeted parenting behaviors that occur soon after birth (Bakermans-Kranenburg et al., 2003; Jeong et al., 2021). Recently, efforts have been made to modify these programs to begin during the prenatal period and focus on preparing parents to sensitively respond to their infants' eventual cry signals (e.g., Labella et al., 2021). Our findings lend support to this idea and provide much needed evidence that intervention efforts can effectively begin prior to the birth of the child, thereby elongating the potential window for intervention opportunities.

## References

- Ablow, J. C., Marks, A. K., Shirley Feldman, S., & Huffman, L. C. (2013). Associations between first-time expectant women's representations of attachment and their physiological reactivity to infant cry. *Child Development*, 84(4), 1373–1391. https://doi.org/10.1111/cdev.12135
- Ainsworth, M. D. S., Blehar, M. C., Waters, E., & Wall, S. (1978). Patterns of attachment. Lawrence Erlbaum Associates.
- Bakermans-Kranenburg, M. J., Van Ijzendoorn, M. H., & Juffer, F. (2003). Less is more: Meta-analyses of sensitivity and attachment interventions in early childhood. *Psychological Bulletin*, 129(2), 195–215. https:// doi.org/10.1037/0033-2909.129.2.195
- Barbosa, M., Beeghly, M., Moreira, J., Tronick, E., & Fuertes, M. (2021). Emerging patterns of infant regulatory behavior in the Still-Face paradigm at 3 and 9 months predict mother–infant attachment at 12 months. *Attachment & Human Development*, 23(6), 814–830. https://doi.org/10 .1080/14616734.2020.1757730
- Beauchaine, T. P., Bell, Z., Knapton, E., McDonough-Caplan, H., Shader, T., & Zisner, A. (2019). Respiratory sinus arrhythmia reactivity across empirically based structural dimensions of psychopathology: A meta-analysis. *Psychophysiology*, 56(5), Article e13329. https://doi.org/10.1111/psyp .13329
- Bell, S. M., & Ainsworth, M. D. S. (1972). Infant crying and maternal responsiveness. *Child Development*, 43(4), 1171–1190. https://doi.org/ 10.2307/1127506
- Bornstein, M. H., & Tamis-Lemonda, C. S. (1997). Maternal responsiveness and infant mental abilities: Specific predictive relations. *Infant Behavior* and Development, 20(3), 283–296. https://doi.org/10.1016/S0163-6383(97)90001-1
- Bowlby, J. (1969/1982). Attachment. Basic Books.
- Conradt, E., & Ablow, J. (2010). Infant physiological response to the stillface paradigm: Contributions of maternal sensitivity and infants' early regulatory behavior. *Infant Behavior & Development*, 33(3), 251–265. https:// doi.org/10.1016/j.infbeh.2010.01.001
- Dawson, M. E., Schell, A. M., & Filion, D. L. (2017). The electrodermal system. In J. T. Cacioppo, L. G. Tassinary, & G. G. Berntson (Eds.), *Handbook* of psychophysiology (pp. 217–243). Cambridge University Press.
- Del Carmen, R., Pedersen, F. A., Huffman, L. C., & Bryan, Y. E. (1993). Dyadic distress management predicts subsequent security of attachment. *Infant Behavior & Development*, 16(2), 131–147. https://doi.org/10 .1016/0163-6383(93)80014-Y
- Dennis, T. A., Buss, K. A., Hastings, P. D., Bell, M. A., Diaz, A., Adam, E. K., Miskovic, V., Schmidt, L. A., Feldman, R., Katz, L. F., Rigterink, T., Strang, N. M., Hanson, J. L., Pollak, S. D., Dahl, R. E., Silk, J. S., Siegle, G. J., Beauchaine, T. P., Cicchetti, D., ... Gates, K. M. (2012). Physiological measures of emotion from a developmental perspective: State of the science. *Monographs of the Society for Research in Child Development*, 77(2), i-viii–i-204. https://doi.org/10.1111/j.1540-5834.2011.00653.x
- DePasquale, C. (2020). A systematic review of caregiver–child physiological synchrony across systems: Associations with behavior and child functioning. *Development and Psychopathology*, 32(5), 1754–1777. https:// doi.org/10.1017/S0954579420001236
- Emery, H. T., McElwain, N. L., Groh, A. M., Haydon, K. C., & Roisman, G. I. (2014). Maternal dispositional empathy and electrodermal reactivity: Interactive contributions to maternal sensitivity with toddler-aged

children. Journal of Family Psychology, 28(4), 505–515. https://doi.org/ 10.1037/a0036986

- Esposito, H., & Bornstein, M. H. (2019). Parenting and infant cry. *Parenting*, 19(1–2), 1–4. https://doi.org/10.1080/15295192.2019.1555415
- Fowles, D. C. (1980). The three arousal model: Implications of Gray's twofactor learning model for heart rate, electrodermal activity, and psychopathy. *Psychophysiology*, *17*(2), 87–104. https://doi.org/10.1111/j.1469-8986.1980.tb00117.x
- Fowles, D. C. (1988). Psychophysiology and psychopathology: A motivational approach. *Psychophysiology*, 25(4), 373–391. https://doi.org/10 .1111/j.1469-8986.1988.tb01873.x
- Gao, M., Speck, B., Ostlund, B., Neff, D., Shakiba, N., Vlisides-Henry, R. D., Kaliush, P. R., Molina, N. C., Thomas, L., Raby, K. L., Crowell, S. E., & Conradt, E. (2022). Developmental foundations of physiological dynamics among mother–infant dyads: The role of newborn neurobehavior. *Child Development*, 93(4), 1090–1105. https://doi.org/10.1111/cdev.13769
- Gratz, K. L., & Roemer, L. (2004). Multidimensional assessment of emotion regulation and dysregulation: Development, factor structure, and initial validation of the difficulties in emotion regulation scale. *Journal of Psychopathology and Behavioral Assessment*, 26(1), 41–54. https:// doi.org/10.1023/B:JOBA.0000007455.08539.94
- Groh, A. M., Propper, C., Mills-Koonce, R., Moore, G. A., Calkins, S., & Cox, M. (2019). Mothers' physiological and affective responding to infant distress: Unique antecedents of avoidant and resistant attachments. *Child Development*, 90(2), 489–505. https://doi.org/10.1111/cdev.12912
- Haley, D. W., & Stansbury, K. (2003). Infant stress and parent responsiveness: Regulation of physiology and behavior during still-face and reunion. *Child Development*, 74(5), 1534–1546. https://doi.org/10.1111/1467-8624.00621
- Hill-Soderlund, A. L., Mills-Koonce, W. R., Propper, C., Calkins, S. D., Granger, D. A., Moore, G. A., Gariepy, J.-L., & Cox, M. J. (2008). Parasympathetic and sympathetic responses to the strange situation in infants and mothers from avoidant and securely attached dyads. *Developmental Psychobiology*, 50(4), 361–376. https://doi.org/10.1002/ dev.20302
- Hout, M., Smith, T. W., & Marsden, P. V. (2016). Prestige and socioeconomic scores for the 2010 census codes (GSS Methodological Report, No. 124, pp. 1–18). General Social Survey, National Opinion Research Center, University of Chicago.
- Jeong, J., Franchett, E. E., de Oliveira, R., Rehmani, C. V., & Yousafzai, K., & K, A. (2021). Parenting interventions to promote early child development in the first three years of life: A global systematic review and metaanalysis. *PLoS Medicine*, 18(5), Article e1003602. https://doi.org/10 .1371/journal.pmed.1003602
- Jones-Mason, K., Alkon, A., Coccia, M., & Bush, N. R. (2018). Autonomic nervous system functioning assessed during the still-face paradigm: A meta-analysis and systematic review of methods, approach and findings. *Developmental Review*, 50(Part B), 113–139. https://doi.org/10.1016/j.dr .2018.06.002
- Joosen, K. J., Mesman, J., Bakermans-Kranenburg, M. J., Pieper, S., Zeskind, P. S., & Van IJzendoorn, M. H. (2013). Physiological reactivity to infant crying and observed maternal sensitivity. *Infancy*, 18(3), 414– 431. https://doi.org/10.1111/j.1532-7078.2012.00122.x
- Joosen, K. J., Mesman, J., Bakermans-Kranenburg, M. J., & van IJzendoorn, M. H. (2013). Maternal overreactive sympathetic nervous system responses to repeated infant crying predicts risk for impulsive harsh discipline of infants. *Child Maltreatment*, 18(4), 252–263. https://doi.org/10 .1177/1077559513494762
- Kirschbaum, C., Pirke, K. M., & Hellhammer, D. H. (1993). The 'Trier Social Stress Test'—A tool for investigating psychobiological stress responses in a laboratory setting. *Neuropsychobiology*, 28(1–2), 76–81. https://doi.org/10.1159/000119004
- Kogan, N., & Carter, A. S. (1996). Mother-infant reengagement following the still-face: The role of maternal emotional unavailability and infant

affect regulation. *Infant Behavior and Development*, *19*(3), 359–370. https://doi.org/10.1016/S0163-6383(96)90034-X

- Labella, M. H., Eiden, R. D., Roben, C. K. P., & Dozier, M. (2021). Adapting an evidence-based home visiting intervention for mothers with opioid dependence: Modified attachment and biobehavioral catch-up. *Frontiers* in Psychology, 12(3332), Article 675866. https://doi.org/10.3389/fpsyg .2021.675866
- Lee, A. H., Kuzava, S., & Bernard, K. (2022). Sociodemographic risk and infants' emerging language ability: Examining the indirect effects of maternal sensitivity and nurturance to distress. *Parenting*, 22(1), 40–60. https://doi.org/10.1080/15295192.2020.1748485
- Leerkes, E. M. (2011). Maternal sensitivity during distressing tasks: A unique predictor of attachment security. *Infant Behavior and Development*, 34(3), 443–446. https://doi.org/10.1016/j.infbeh.2011.04.006
- Leerkes, E. M., Blankson, A. N., & O'Brien, M. (2009). Differential effects of maternal sensitivity to infant distress and nondistress on socialemotional functioning. *Child Development*, 80(3), 762–775. https:// doi.org/10.1111/j.1467-8624.2009.01296.x
- Leerkes, E. M., Gedaly, L., & Su, J. (2016). Parental sensitivity and infant attachment. In L. Balter & C. S. Tamis-LeMonda (Eds.), *Child psychology: A handbook of contemporary issue* (3rd ed., pp. 21–41). Taylor & Francis.
- Leerkes, E. M., Girod, S. A., Buehler, C., Shriver, L. H., & Wideman, L. (2023). Interactive effects of maternal physiological arousal and regulation on maternal sensitivity: Replication and extension in an independent sample. *Developmental Psychobiology*, 65(2), Article e22375. https://doi.org/ 10.1002/dev.22375
- Leerkes, E. M., Parade, S. H., & Gudmundson, J. A. (2011). Mothers' emotional reactions to crying pose risk for subsequent attachment insecurity. *Journal of Family Psychology*, 25(5), 635–643. https://doi.org/10.1037/ a0023654
- Leerkes, E. M., Su, J., Calkins, S. D., O'Brien, M., & Supple, A. (2017). Maternal physiological dysregulation while parenting poses risk for infant attachment disorganization and behavior problems. *Development and Psychopathology*, 29(1), 245–257. https://doi.org/10.1017/S0954579416000122
- Leerkes, E. M., Su, J., Calkins, S. D., Supple, A. J., & O'Brien, M. (2016). Pathways by which mothers' physiological arousal and regulation while caregiving predict sensitivity to infant distress. *Journal of Family Psychology*, 30(7), 769–779. https://doi.org/10.1037/fam0000185
- Leerkes, E. M., Supple, A. J., O'Brien, M., Calkins, S. D., Haltigan, J. D., Wong, M. S., & Fortuna, K. (2015). Antecedents of maternal sensitivity during distressing tasks: Integrating attachment, social information processing, and psychobiological perspectives. *Child Development*, 86(1), 94–111. https://doi.org/10.1111/cdev.12288
- Leerkes, E. M., & Zhou, N. (2018). Maternal sensitivity to distress and attachment outcomes: Interactions with sensitivity to nondistress and infant temperament. *Journal of Family Psychology*, 32(6), 753–761. https://doi.org/10.1037/fam0000420
- McElwain, N. L., & Booth-LaForce, C. (2006). Maternal sensitivity to infant distress and nondistress as predictors of infant–mother attachment security. *Journal of Family Psychology*, 20(2), 247–255. https://doi.org/10.1037/ 0893-3200.20.2.247
- Mesman, J., Linting, M., Joosen, K. J., Bakermans-Kranenburg, M. J., & Van IJzendoorn, M. H. (2013). Robust patterns and individual variations: Stability and predictors of infant behavior in the still-face paradigm. *Infant Behavior & Development*, 36(4), 587–598. https://doi.org/10 .1016/j.infbeh.2013.06.004
- Mesman, J., Van IJzendoorn, M. H., & Bakermans-Kranenburg, M. J. (2009). The many faces of the Still-Face Paradigm: A review and metaanalysis. *Developmental Review*, 29(2), 120–162. https://doi.org/10 .1016/j.dr.2009.02.001
- Mills-Koonce, W. R., Gariepy, J. L., Propper, C., Sutton, K., Calkins, S., Moore, G., & Cox, M. (2007). Infant and parent factors associated with early maternal sensitivity: A caregiver-attachment systems approach.

Infant Behavior and Development, 30(1), 114–126. https://doi.org/10 .1016/j.infbeh.2006.11.010

- MindWare Technologies. (2014). Heart Rate Variability (HRV): User reference guide. http://downloads.mindwaretech.com/HRV/HRV%203.1.0-2%20User%20Guide.pdf
- Moore, G. A., Hill-Soderlund, A. L., Propper, C. B., Calkins, S. D., Mills-Koonce, W. R., & Cox, M. J. (2009). Mother–infant vagal regulation in the face-to-face Still-Face Paradigm is moderated by maternal sensitivity. *Child Development*, 80(1), 209–223. https://doi.org/10.1111/j.1467-8624.2008.01255.x
- Muthén, L. K., & Muthén, B. O. (2017). Mplus user's guide (8th ed.).
- NICHD Early Child Care Research Network. (1996). Characteristics of infant child care: Factors contributing to positive caregiving. *Early Childhood Research Quarterly*, 11(3), 269–306. https://doi.org/10.1016/S0885-2006(96)90009-5
- Ostlund, B. D., Measelle, J. R., Laurent, H. K., Conradt, E., & Ablow, J. C. (2017). Shaping emotion regulation: Attunement, symptomatology, and stress recovery within mother–infant dyads. *Developmental Psychobiology*, 59(1), 15–25. https://doi.org/10.1002/dev.21448
- Porges, S. W. (2007). The polyvagal perspective. *Biological Psychology*, 74(2), 116–143. https://doi.org/10.1016/j.biopsycho.2006.06.009
- Raby, K. L., Roisman, G. I., Simpson, J. A., Collins, W. A., & Steele, R. D. (2015). Greater maternal insensitivity in childhood predicts greater electrodermal reactivity during conflict discussions with romantic partners in adulthood. *Psychological Science*, 26(3), 348–353. https://doi.org/10 .1177/0956797614563340
- Rosenblum, K. L., McDonough, S., Muzik, M., Miller, A., & Sameroff, A. (2002). Maternal representations of the infant: Associations with infant response to the Still Face. *Child Development*, 73(4), 999–1015. https:// doi.org/10.1111/1467-8624.00453
- Semega, J., & Kollar, M. (2022). Income in the United States: 2021 (P60-273). U.S. Census Bureau, Current Population Reports, U.S. Government Publishing Office.
- Shaffer, F., McCraty, R., & Zerr, C. L. (2014). A healthy heart is not a metronome: An integrative review of the heart's anatomy and heart rate variability. *Frontiers in Psychology*, 5, Article 1040. https://doi.org/10.3389/ fpsyg.2014.01040

- Speck, B., Raby, L., Conradt, E., & Crowell, S. (2023, January 18). Prenatal predictors of the emergence of infant attachment. https://doi.org/10 .17605/OSF.IO/EMFP5
- Sroufe, L. A. (1997). Emotional development: The organization of emotional life in the early years. Cambridge University Press.
- Thompson, R. A. (1997). Sensitivity and security: New questions to ponder. Child Development, 68(4), 595–597. https://doi.org/10.2307/1132109
- Tronick, E., Als, H., Adamson, L., Wise, S., & Brazelton, T. B. (1978). The infant's response to entrapment between contradictory messages in face-to-face interaction. *Journal of the American Academy of Child Psychiatry*, 17(1), 1–13. https://doi.org/10.1016/S0002-7138(09)62273-1
- U.S. Census Bureau. (2021). *QuickFacts Salt Lake County, Utah.* U.S. Department of Commerce. Retrieved May 15, 2023, from https://data.census.gov/
- U.S. Census Bureau. (2022). QuickFacts United States. U.S. Department of Commerce. Retrieved June 21, 2023, from https://data.census.gov/
- Vermillet, A.-Q., Tølbøll, K., Litsis Mizan, S., Skewes, J., & Parsons, C. E. (2022). Crying in the first 12 months of life: A systematic review and metaanalysis of cross-country parent-reported data and modeling of the "cry curve". *Child Development*, 93(4), 1201–1222. https://doi.org/10.1111/ cdev.13760
- Whisman, M. A., & McClelland, G. H. (2005). Designing, testing, and interpreting interactions and moderator effects in family research. *Journal of Family Psychology*, 19(1), 111–120. https://doi.org/10.1037/0893-3200 .19.1.111
- Wright, N., Hill, J., Sharp, H., & Pickles, A. (2018). Maternal sensitivity to distress, attachment and the development of callous-unemotional traits in young children. *Journal of Child Psychology and Psychiatry*, 59(7), 790–800. https://doi.org/10.1111/jcpp.12867
- Xu, N., & Groh, A. M. (2023). The significance of mothers' attachment representations for vagal responding during interactions with infants. *Attachment & Human Development*, 25(1), 50–70. https://doi.org/10 .1080/14616734.2021.1876615

Received January 18, 2023 Revision received July 5, 2023

Accepted August 3, 2023 ■